

Author's response to reviews

Title: Implementing the WHO integrated tool to assess quality of care for mothers, newborns and children: results and lessons from five districts in Malawi

Authors:

Helen Smith (helen.smith@lstmed.ac.uk)

Atnafu Getachew Asfaw (agasfaw@unicef.org)

Kyaw Myint Aung (kaung@unicef.org)

Lastone Chikoti (chikotilas@yahoo.com)

Florence Mgawadere (Florence.Mgawadere@lstmed.ac.uk)

Luigi D'Aquino (Luigi.DAquino@lstmed.ac.uk)

Nynke van den Broek (Nynke.vandenBroek@lstmed.ac.uk)

Version: 4 Date: 31 Jul 2017

Author's response to reviews:

1. In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Thank you. We have added the following information in the above-mentioned section: "Two authors (AGA and KMA) are employees of UNICEF and provided technical oversight of data collection and contributed to writing the manuscript. The content represents the authors' views and not the official views of UNICEF or WHO."

2. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided before the declarations section.

We have added a list of abbreviations before the declarations section.

3. On the title page please include email addresses for all listed authors.

We have added email addresses for all listed authors.

4. The statement "All the data collected as part of this assessment is presented in this paper and can be found in Table 1." is invalid. This is because Availability of Data and Materials refers to the raw data-set upon which the summary statistics in the manuscript are based.

Please detail where the data supporting your findings can be found - including information in any supplementary files. If you do not wish to share your data, please state that data will not be shared, together with a valid reason.

For your information: Here are some examples of Availability of Data and Materials statements that you may choose to adapt:

- The datasets generated during and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]
- The datasets generated and/or analysed during the current study are available from the corresponding author on reasonable request.
- The datasets generated and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.
- The datasets generated and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC].

Thanks for pointing this out. We have now revised the sentence under ‘availability of data’ to read: “The data generated and/or analysed for this study are available from the corresponding author on reasonable request”.

5. Please remove the file “Response to Reviewers 16th June” as it is no longer required at this stage in the editorial process.

This file has been removed from the online submission area.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript is now a single, clean, final version. The tables are also clean and final versions and figure 1 has been uploaded as a separate file.